Author's response to reviews

Title: Use of information and communication technologies to support effective work practice innovation in the health sector: a multi-site study

Authors:

Johanna I Westbrook (J.Westbrook@usyd.edu.au)
Jeffrey Braithwaite (J.Braithwaite@unsw.edu.au)
Kathryn Gibson (Kathy.Gibson@sswahs.nsw.gov.au)
Richard Paoloni (rpao@bigpond.net.au)
Joanne L Callen (J.Callen@usyd.edu.au)
Andrew Georgiou (A.Georgiou@usyd.edu.au)
Nerida Creswick (N.Creswick@usyd.edu.au)
Louise Robertson (Louise.Robertson@sswahs.nsw.gov.au)

Version: 3 Date: 20 October 2009

Author's response to reviews:

Dear Nina

In the revised version we have added two paragraphs where we outline two specific substudies as exemplars of the types of types which will be included in this large multi-site study. In these examples we have provided indications of the sample sizes and compositions as requested.

Regards
Johanna Westbrook